Author’s response to reviews

Title: Is mitral valve repair superior to replacement for chronic ischemic mitral regurgitation with left ventricular dysfunction?

Authors:

Zhibing Qiu (qiuzhibingjs@yahoo.com.cn)
Xin Chen (stevecx@sina.com)
Ming Xu (xuming2008@163.com)
Yingshuo Jiang (jiangyinsuo@sina.com)
Liqiong Xiao (xiaoliqiong@163.com)
LeLe Liu (liulele1978@yahoo.com.cn)
Liming Wang (wlm1971@yahoo.com.cn)

Version: 3 Date: 29 October 2010

Author's response to reviews: see over
Dear Editors,

Thank you for your letter of October 28, enclosing the editors’ comments. We are grateful for formatting changes required to my manuscript (MS: 1110237170378449).

According to comments from the independent editorial production team, we had revised the de novo manuscript. Now I submit the revised manuscript to you, enclosing the point-to-point description of the changes made. I sincerely hope that you would be satisfied with the manuscript, and your advice would be honored.

I am looking forward to hearing from you soon. Should you have any further inquiries, please do not hesitate to contact me.

Yours faithfully,

Xin Chen, M.D. & Ph.D.
Professor

Address for correspondence:
Prof. Xin Chen,
Dept Cardiothoracic and vascular Surgery, Nanjing First Hospital affiliated to Nanjing Medical University,
68 Changle Rd, 210006 Nanjing, China.
E-mail: stevecx@sina.com
Tel&fax: 86-25-52247821, 13605197742 (Mobile Phone)
Response to formatting changes

Thank you very much for your kind comments.

Abstract: I am sure that the abstract is identical in the manuscript and the submission system.

Tables: I have added the tables as additional files in the manuscript following the references. and including the figures as an additional files section in the manuscript.

References - I have replaced consecutive numbers in square brackets according to the instruction, which were highlighted with red colour text.

Title - I have changed the title formatting according to the instruction.